Author’s response to reviews

Title: Enteral vitamin A for reducing severity of bronchopulmonary dysplasia in extremely preterm infants: a randomised controlled trial

Authors:
Abhijeet Rakshasbhuvankar (abhijeet.rakshasbhuvankar@health.wa.gov.au)
Sanjay Patole (sanjay.patole@health.wa.gov.au)
Karen Simmer (karen.simmer@health.wa.gov.au)
Jane Pillow (jane.pillow@uwa.edu.au)

Version: 1 Date: 02 Oct 2016

Author’s response to reviews:

2nd October 2016

The Editor,
BMC Pediatrics

Subject: Oral (enteral) vitamin A for prevention of bronchopulmonary dysplasia in extremely preterm infants: a randomised controlled trial" (BPED-D-16-00500)

Dear Editor

Please find below the response to the points raised in your correspondence dated 23/09/2016.

1. Ethical and funding approval documentation: I have sent the documents to the email address as requested.
2. Funding: The study is externally funded and has received funding from Channel 7 Telethon Trust. However, the current study protocol has not been peer-reviewed by the funding body.

3. Study status: We plan to start recruitment from 1st November 2016. We are not recruiting at the moment.

4. Related articles: I confirm that there are no publications containing the results of this study have already been published or submitted to any journal.

We look forward to hearing from you.

Thanking you

Yours faithfully

Dr Abhijeet Rakshashbhuvankar
Clin Prof Sanjay Patole
W/Prof Karen Simmer
A/Prof Jane Pillow